Author's response to reviews

Title: Interferon beta-1a for the maintenance of remission in patients with Crohn's disease: results of a phase II dose-finding study

Authors:

Claudia Pena Rossi (claudia.penarossi@merckserono.net)
Stephen B Hanauer (shanauer@medicinebsd.uchicago.edu)
Ratko Tomasevic (tomasnik@eunet.yu)
John O Hunter (john.hunter@addenbrookes.nhs.uk)
Ira Shafran (ira@shafran.net)
Hans Graffner (hans.graffner@merckserono.net)

Version: 7 Date: 16 March 2009

Author's response to reviews: see over
Melissa Norton, MD  
Editor-in-Chief  
BMC Gastroenterology

16 March 2009

Dear Dr Norton

Re: Interferon beta-1a for the maintenance of remission in patients with Crohn’s disease: results of a phase II dose-finding study

We thank the journal for the comments sent and have amended the text accordingly.

1. All author names are now consistent between the manuscript and the submission system.
2. The text has been added to the authors’ contribution section.
3. The Ethical approval details have been removed from the Appendix (which has been deleted) and is now in a separate section within the Methods.
4. The references have been updated as requested.
5. Research Article has been removed from the top of the manuscript.
6. Figure 2 legend has been amended as requested.
7. Additional, minor amendments have been made to the text.

We hope that it is now acceptable for publication.

Yours sincerely

Claudia Pena Rossi  
Merck Serono S.A. – Geneva  
15bis Chemin des Mines  
CH-1211 Geneva  
Switzerland  
Tel.: +41 22 414 3834  
Fax: +41 22 414 3012  
Email: claudia.penarossi@merckserono.net